\begin{table}%t2
%\centering
\par
\caption{\label{tab:entries}Additional SPECFIND~V2.0 catalogue entries.}
\begin{tabular}{lcrrrrcl}
\hline\hline\noalign{\smallskip}
Catalog & I/S\tablefootmark{1} & Frequency & Resolution & $S_{\rm min}$ & Number of & Percentage\tablefootmark{2} &  Reference \\
name &  & (MHz) & (arcmin) & (mJy) & sources &  & \\
\hline
NEK  &   I &     31 &  12.0  & 2000   &   703 &   4  &  Kassim (\cite{Kassim1988}) \\
8C   &   I  &    38  &  4.5   & 700   &  5859 &  66  &  Hales et~al. (\cite{Hales1995}) \\
CRJ2004& I  &    74  &  0.42  & 150   &   949 &  63  &  Cohen et~al. (\cite{Cohen2004}) \\
VLSS  &  I &     74  &  1.33  & 400   & 68~308 &  66 &  Cohen et~al. (\cite{Cohen2007}) \\
TRC2006b&I &     74 &   0.5   &   0.3 &   725 &   0.2  &  Tasse et~al. (\cite{Tasse2006}) \\
Cula  &  I &     80  &  3.7  & 2000   &  2173 &  49  &  Slee (\cite{Slee1995}) \\
6C  &    I  &   151  &  4.2   & 300   & 27~666 &  84  & Baldwin et~al. (\cite{Baldwin1985}) \\ 
& & & & & & & Hales et~al. (1988/80/91/93) \\
7C  &    I  &   151  &  1.17  & 200   & 43~683 &  77  &  Hales et~al. (\cite{Hales2007}) \\
    Culb  &  I &    160  &  1.85 & 1000   &  2042 &  66  &  Slee (\cite{Slee1995}) \\
    TRB2007a&I  &   240  &  0.25  &   6   &   466 &  63  &  Tasse et~al. (\cite{Tasse2007}) \\
    TRC2006a&I &    325  &  0.12  &   2.5 &   843 &  47  &  Tasse et~al. (\cite{Tasse2006}) \\
    WSTB90&  I  &   327  &  1.5   &   3   &   407 &      &  Oort et~al. (\cite{Oort1988}) \\
    W93a  &  I &    327  &  1.0   &   1   &  4157 &  32  &  Wieringa (\cite{Wieringa1993}) \\
    WSTB2 &  I &    327  &  1.5  &   10   &   309 &      &  Righetti et~al. (\cite{Righetti1988}) \\
    WSRTGP & I &    327 &   1.0   &   5   &  3984 &  14  &  Taylor et~al. (\cite{Taylor1996}) \\
    PSRa  &    &    400 &   1.0   &   0.1 &   561 &   0.4  &  Taylor et~al. (\cite{Taylor1993}) \\
    32P   &  I  &   408  &  4.0   &  30   &   494 &  61  &  Leahy \& Roger (\cite{Leahy1996}) \\
    5C5  &   I &    408  &  1.5   &  10   &   214 &  60  &  Pearson (1975) \\
    5C6   &  I &    408  &  1.5   &   6   &   267 &  68  &  Pearson (\cite{Pearson1978}) \\
    5C7   &  I  &   408  &  1.5   &  10   &   235 &  41  &  Pearson (\cite{Pearson1978}) \\
    5C12  &  I &    408  &  1.5  &    2   &   308 &  70  &  Benn et~al. (\cite{Benn1982}) \\
    5C13 &   I &    408  &  1.5   &  12   &   238 &  60  &  Benn (\cite{Benn1995}) \\
    51P  &   I &    408  &  4.0   &  80   &   383 &  31  &  Green \& Riley (\cite{Green1995}) \\
    5C12a &  I &    408  &  1.33 &    5   &   680 &  49  &  Benn \& Kenderdine (\cite{Benn1991}) \\
    DRAOP &  I &    408  &  3.5   &       &   915 &  62  &  Landecker \& Caswell (\cite{Landecker1983})\\
          &    &         &        &       &       &      &  VizieR VIII/55$^{3}$\\
    CGPSEa & I &    408  &  3.0   &  10   &   140 &  24  &  Kerton et~al. (\cite{Kerton2007}) \\
    PSRb  &   &     600  &  1.0   &   0.4 &   352 &   0.4  &  Taylor et~al. (\cite{Taylor1993}) \\
    W93b  &  I &    608  &  0.5   &   1   &  1693 &  62  &  Wieringa (\cite{Wieringa1993}) \\
    TRB2007b&I  &   610  &  0.11  &   1  &   1037 &  35  &  Tasse et~al. (\cite{Tasse2007}) \\
    FLSGMRT &I  &   610  &  0.1   &   0.1 &  3944 &   1  &  Garn et~al. (\cite{Garn2007}) \\
    NAIC  &  S &    611  & 12.6   & 350   &  3122 &  45  &  Durdin et~al. (\cite{Durdin1975}) \\
    MOST  &  I &    843  &  0.733 &  40   &   348 &  59  &  Jones \& McAdam (\cite{Jones1992}) \\
    MGPS2  & I  &   843  &  0.75  &  10  &  48~850  &  3  &  Murphy et~al. (\cite{Murphy2007}) \\
    ATESP  & I &   1400  &  0.23  &   0.3 &  3370 &   1  &  Wieringa \& Ekers (\cite{Wieringa2000}) \\
    PSRc &     &   1400  &  1.0   &   0.1 &   445 &   0.4  &  Taylor et~al. (\cite{Taylor1993}) \\
    OWH82 &  S &   1400  & 10.0   & 100   &   487 &  63  &  Owen et~al. (\cite{Owen1982}) \\
    GPSR  &  I &   1400  &  0.083 &   5   &  1992 &  11  &  Zoonematkermani et~al. (\cite{Zoonematkermani1990}) \\
    LO95  &  I  &  1400  &  0.25  &  10   &   375 &      &  Ledlow \& Owen (\cite{Ledlow1995}) \\
    PDF   &  I &   1400  &  0.15  &   0.1 &  1079 &   0.1  &  Hopkins et~al. (\cite{Hopkins1998}) \\
    VIRMOS&  I &   1400  &  0.1  &    0.1 &  1103 &   4  &  Bondi et~al. (\cite{Bondi2003}) \\
    FHW95a & S &   1400  & 15.2   &  40   &   192 &  45  &  Filipovic et~al. (\cite{Filipovic1995}) \\
    WST32 &  I &   1400  &  1.2  &   10   &   215 &  29  &  Fanti et~al. (\cite{Fanti1981}) \\
    37W   &  I &   1400  &  0.6  &    1   &    53 &  28  &  Walterbos et~al. (\cite{Walterbos1985}) \\
    ELAISR & I &   1400  &  0.25  &   0.1 &   965 &   1  &  Ciliegi et~al. (\cite{Ciliegi1999}) \\
    MGC2004& I &   1400  &  0.23  &   0.03&  1048 &   1  &  Morganti et~al. (\cite{Morganti2004}) \\
    WBH2005 &I &   1400  &  0.1   &   2  &   6919  &  4  &  White et~al. (\cite{White2005}) \\
    RRF1  &  S  &  1410  &  9.4   & 100   &   884 &      &  Reich et~al. (\cite{Reich1990}) \\
    RRF   &  S &   1410  &  9.4   &  80   &  1830 &      &  Reich et~al. (\cite{Reich1997}) \\
    QC93  &  S &   1410 &  10.0  &  400   &   171 &  57  &  Quiniento \& Cersosimo (\cite{Quiniento1993}) \\
    CCH85  & I &   1411 &   0.33 &    5   &   208 &   9  &  Coleman et~al. (\cite{Coleman1985}) \\
    WSTB &   I &   1412 &   0.385 &   1   &   536 &      &  Windhorst et~al. (\cite{Windhorst1984}) \\
    WSTB1 &  I &   1412  &  0.385 &   1   &   359 &      &  Oort (\cite{Oort1987}) \\
    33P    & I  &  1420  &  1.0   &   3   &   255 &  36  &  Leahy \& Roger (\cite{Leahy1996}) \\
\hline
    FBR2002& I &   1420  &  1.633 &   3   &   534 &  43  &  Filipovic et~al. (\cite{Filipovic2002}) \\
    CGPSEb & I  &  1420 &   1.0   &   3  &    140  & 27  &  Kerton et~al. (\cite{Kerton2007}) \\
    RLM94  & I  &  1465  &  0.075 &   5   &   725 &  42  &  Roettgering et~al. (\cite{Roettgering1994}) \\
    P82\_1 &  I  &  1465  &  0.43  & 200   &   404 &  71  &  Perley (\cite{Perley1982}) \\ 
    NEP  &   I  &  1490  &  0.333 &   1   &  2435 &  16  &  Kollgaard et~al. (\cite{Kollgaard1994}) \\
    DC78 &   S  &  2370  &  2.7   &   1   &   858 &   29  &  Dressel \& Condon (\cite{Dressel1978}) \\
    FBR2002a&I &   2370  &  0.667 &   1   &   697 &  33  &  Filipovic et~al. (\cite{Filipovic2002}) \\
    FHW95b & S &   2450  &  8.85 &   30   &   334 &  54  &  Filipovic et~al. (\cite{Filipovic1995}) \\
    FORa  &  S &   2695  &  4.78  &  20   &   221 &  81  &  Forkert \& Altschuler (\cite{Forkert1987}) \\
    RFS  &   S  &  2695  &  4.9   &  30   &  1212 &  27  &  Reich et~al. (\cite{Reich1984}) \\
    RGB11 &  S  &  2700  &  4.3   &  50   &   697 &  50  &  Reich et~al. (\cite{Reich2000}) \\
    PBD2003& S  &  2700  &  8.0   & 100   &  1432 &  9  &  Paladini et~al. (\cite{Paladini2003}) \\
    KMP90 &  S  &  4730  &  2.8   &  15   &   752 &  85  &  Kulkarni et~al. (\cite{Kulkarni1990}) \\
    FORb  &  S &   4750  &  2.71  &  25   &   227 &  85  &  Forkert \& Altschuler (\cite{Forkert1987}) \\
    FHW95c & S &   4750  &  4.8  &   15   &   368 &  53  &  Filipovic et~al. (\cite{Filipovic1995}) \\
    NAICGB & S &   4775  &  2.8   &   8  &   2453 &  65  &  Lawrence et~al. (\cite{Lawrence1983}) \\
    A86    & S &   4760  &  2.8  &   15   &   882 &  74  &  Altschuler (\cite{Altschuler1986}) \\
    FBR2002b&I &   4800 &   0.5   &   1   &    75 &  40  &  Filipovic et~al. (\cite{Filipovic2002}) \\
    GDP   &  I &   4850  &  0.5  &    0.3 &   253 &   8  &  Gregorini et~al. (\cite{Gregorini1994}) \\
    CAB95 &  S  &  4850  &  3.5   &  25   &   351 &  76  &  Condon et~al. (\cite{Condon1995}) \\
    JRB99  & I &   4860  &  0.03 &    1   &   298 &  28  &  Jackson et~al. (\cite{Jackson1999}) \\
    FPD2001a&I &   4860  &  6.7e-3&   1   &   213 &  36  &  Fanti et~al. (\cite{Fanti2001}) \\
    ADP79  & S &   4875  &  2.6   & 100   &   569 &  19  &  Altenhoff et~al. (\cite{Altenhoff1979}) \\
    RGB   &  I  &  4885  &  0.067 &  13   &  1861 &  56  &  Laurent-Muehleisen et~al. (\cite{Laurent1997}) \\
    P82\_2  & I  &  4885  &  0.1   & 200   &   404 &  82  &  Perley (\cite{Perley1982}) \\ 
    KR    &  I &   4890  &  0.07 &   10   &   195 &  41  &  Fich (\cite{Fich1986}) \\
    GPSR5 &  I  &  4900  &  0.067 &   1   &  1286 &   3  &  Becker et~al. (\cite{Becker1994}) \\
    Slee  &  I &   4900  &  0.67  &  0.2  &   177 &  21  &  Slee et~al. (1998) \\
    HCS79 &    &   5000  &  0.54  & 240   &   702 &   3  &  Haynes et~al. (\cite{Haynes1979}) \\
    RGB6  &  S  &  5000  &  2.4   &  40   &   729 &  56  &  Reich et~al. (\cite{Reich2000}) \\
    GPA2  &  S  &  8350  &  9.7   & 900   &   555 &   6  &  Langston et~al. (\cite{Langston2000}) \\
    CLASS &  I &   8400  &  5.5e-3&   0.1 & 21486 &  39  &  Myers et~al. (\cite{Myers2003}) \\
    FPD2001b&I &   8460  &  3.3e-3&   0.2&    199 &  33  &  Fanti et~al. (\cite{Fanti2001}) \\
    FHW95d&  S &   8550  &  2.7  &   20   &   205 &  47  &  Filipovic et~al. (\cite{Filipovic1995}) \\
    FBR2002c&I &   8640  &  0.27  &   1   &    54 &  41  &  Filipovic et~al. (\cite{Filipovic2002}) \\
    NKB95 &  S &  10~550  &  1.15 &    3   &   202 &  57  &  Niklas et~al. (\cite{Niklas1995}) \\
    B3VLA &  S  & 10~600  &  1.15  &  10   &   981 &  80  &  Gregorini et~al. (\cite{Gregorini1998}) \\    
    RGB2  &  S  & 10~700  &  1.2   &  30   &   698 &  53  &  Reich et~al. (\cite{Reich2000}) \\
    GPA1 &   S &  14~350  &  6.6   &2000   &   365 &   7  &  Langston et~al. (\cite{Langston2000}) \\
9C    &  I &  15~000  &  0.42  &  25   &   242 &  71  &  Waldram et~al. (\cite{Waldram2003}) \\
\hline
\end{tabular}
\tablefoot{\tablefoottext{1}{S: single dish; I: interferometer.} 
\tablefoottext{2}{Percentage of sources with identified spectrum. Only unambiguous source names could be counted.}
\tablefoottext{3}{This catalogue is a compilation of tables in 27~articles. Landecker \& Caswell (1983) is the 
first reference.}}
\end{table}